# Adherence to clinical guidelines for monitoring diabetes in primary care settings

Mingliang Dai<sup>1</sup>, Michael R. Peabody<sup>1</sup>, Lars E. Peterson<sup>1</sup>, Arch G. Mainous III<sup>2,3</sup>

## Abstract

**Objective:** Adherence to clinical guidelines is key to improving diabetes care. Contemporary knowledge of guideline adherence is lacking. This study sought to produce a national snapshot of primary care physicians' (PCPs) adherence to the American Diabetes Association guidelines for monitoring diabetes and determine whether continuity of care promotes adherence.

**Methods:** Using the 2013 National Ambulatory Medical Care Survey, we examined adherence to ordering hemoglobin  $A_{1c}$  (Hb $A_{1c}$ ) and lipid profile tests as recommended by the American Diabetes Association for monitoring diabetes in 2379 primary care visits of patient with diabetes.

**Results:** In the preceding 12 months, less than 60.0% of the patients were given a test recommended for monitoring diabetes (58.0% for HbA<sub>1c</sub> and 57.0% for lipid profile). Continuity of care with PCPs increased the odds of adhering to diabetes monitoring guidelines by 36.0% for the HbA<sub>1c</sub> test (P=0.06) and by 76.0% for the lipid profile test (P=0.0006).

**Conclusion:** A substantial gap exists in achieving optimal monitoring for diabetes in primary care settings in the United States. While PCPs are ideally positioned to ensure that guidelines are closely followed, we found that even in primary care settings, patient-provider continuity of care was associated with guideline adherence.

Keywords: Diabetes; guideline adherence; primary care; continuity of care

**Significance statement:** Adherence to clinical guidelines is an integral part of the World Health Organization's Global Action Plan to reduce the impact of diabetes. In this study, we used the 2013 National Ambulatory Care Survey to assess US physicians' adherence to the American Diabetes Association's guidelines for monitoring diabetes. We found a substantial gap between what was done in primary care settings and what is recommended for optimal diabetes management. Our findings support the positive effect of continuity of care on improving diabetes quality of care.

#### Introduction

Diabetes affects more than one in ten adult Americans aged 20 years or older, and its prevalence has increased dramatically over the past two decades and continues to grow disproportionally among Hispanics and non-Hispanic blacks [1–4]. Despite being one of the leading causes of hospitalization in the United States [5], acute and chronic complications of diabetes may be prevented or mitigated with evidence-based therapeutic treatments [6, 7].

Ensuring appropriate quality of care is key to effective management of diabetes. The National Committee for Quality Assurance's 1. American Board of Family Medicine, 1648 McGrathiana Parkway, Suite 550, Lexington, KY 40511, USA

2. University of Florida, Department of Health Services Research, Management and Policy, 1225 Center Drive, HPNP 3107, Gainesville, FL 32611, USA

3. University of Florida, Department of Community Health and Family Medicine, 1225 Center Drive, HPNP 3107, Gainesville, FL 32611, USA

CORRESPONDING AUTHOR: Mingliang Dai, PhD American Board of Family Medicine, 1648 McGrathiana Parkway, Suite 550, Lexington, KY 40511, USA Tel.: +1-859-2595626 x 1347 E-mail: mdai@theabfm.org

Received 28 June 2018; Accepted 3 August 2018

<sup>© 2018</sup> Family Medicine and Community Health. Creative Commons Attribution-NonCommercial 4.0 International License



Healthcare Effectiveness Data and Information Set (HEDIS) measures important dimensions of care and services for diabetes that are used by more than 90.0% of US health plans [8]. The diabetes quality measures follow the clinical guidelines issued by the American Diabetes Association (ADA) [9]. Yet the outcomes of controlling diabetes have been less than optimal. Of the patients receiving diabetic medications, 40.0% did not have their diabetes under control [10]. More specifically, only 29.0%–57.0% of patients reached their goal for glycemic control and 31.0%–46.0% for lipid targets [11–13]. To remedy this deficiency, considerable research has demonstrated that quality improvement strategies are effective in improving clinical outcomes (e.g., glycemic control or reducing hemoglobin A<sub>1c</sub> [HbA<sub>1c</sub>] values) [14–18].

In contrast to the national effort to achieve better treatment goals, contemporary knowledge of physicians' adherence to process measures (e.g., ordering regular HbA<sub>1c</sub> and lipid profile tests), which is equally important for successful long-term diabetes management, is lacking. Studies from 20 years ago found very poor adherence to ADA standards, with only 15.0%-20.0% of diabetic patients receiving recommended tests [19, 20]. Despite increasing attention to quality of care since then, in 2010 only one in four adults aged 40 years or older received recommended tests [21]. In comparison, adherence to guidelines in European countries was much higher, ranging from 65.0% to 85.0% for HbA<sub>1c</sub> tests and from 67.0% to 89.0% for annual lipid profile tests [22–24].

Improving the quality of care for patients with diabetes is crucial in relieving both the disease burden on the patients and the economic burden on society [25]. Given that most patients with diabetes are cared for in primary care settings [26–29], it is important to assess adherence to diabetes monitoring guidelines among primary care physicians (PCPs). Gaps in adherence will alert physicians to missed opportunities that may help achieve optimal long-term care for patients with diabetes.

Moreover, adherence to guidelines needs to be examined in the context of patient-provider continuity relationships, which is a core principle of primary care. A handful of studies that directly assessed the role of patient-provider continuity reported inconsistent results. Patients with diabetes who indicated a usual source of care were significantly more likely receive a diagnosis and have better glycemic control [23, 30]. However, the likelihood of receiving HbA<sub>1c</sub> and lipid profile tests was not statistically different regardless of the level of continuity [31]. These findings, while insightful, were reflective of diabetes care in the 1990s. Research built on current patient encounters is needed to determine the effect of continuity in today's changing primary care settings. This study sought to produce a contemporary national snapshot of PCPs' adherence to diabetes monitoring guidelines and to determine whether continuity of care promotes adherence.

# Methods

# Data

We used the 2013 National Ambulatory Medical Care Survey (NAMCS) public use file to draw a nationally representative sample of primary care visits of patients with diabetes. The NAMCS is a national probability sample survey of patient visits to office-based physicians conducted by the National Center for Health Statistics [32]. The sampling frame for the 2013 NAMCS was composed of all physicians whose names are contained in the American Medical Association and the American Osteopathic Association master files who were office based, principally engaged in patient care activities, not federally employed, not in specialties of anesthesiology, pathology, and radiology, and younger than 85 years at the time of the survey. The 2013 NAMCS included 11,212 physicians: 10,595 doctors of medicine and 617 doctors of osteopathy. Of the 6999 remaining eligible physicians, 2705 participated in the survey, and data were collected for 54,873 patient visits [33].

## Sample

We started with the full sample of 54,873 patient visits in the 2013 NAMCS and applied two sample selection criteria to create the analytical sample. First, we excluded 40,113 visits that were made to non-PCPs, leaving us with 14,760 visits made to PCPs, who we defined as general practitioners, family physicians, and general internists. Second, we excluded 12,381 visits of nondiabetic patients on the basis of both diagnoses and reports by physicians that were used to identify patients with diabetes. An individual was determined to have diabetes if the primary, secondary, or tertiary diagnosis code was in the range of 250.xx, or if the physician indicated that the patient had diabetes regardless of the diagnoses previously entered.

Our analytical sample included 2379 primary care visits of patients with diabetes.

#### Measures

We examined guideline adherence for two common laboratory tests recommended for monitoring diabetes: the HbA1c test and the lipid profile test. We used the 2013 ADA guidelines as the benchmark for adherence. On the basis of expert consensus and clinical experience, the 2013 ADA guidelines recommended that physicians order (1) the HbA1c test twice a year for patients with stable glycemic control and more frequently for those not meeting the goals and (2) the lipid profile test at least annually [9]. As these recommendations are consistent with the preceding ones released in 2012, 2011, and 2010 [34-36], it would be reasonable to expect that clinicians' practice in 2013 reliably reflected their adherence to these guidelines. Since the number of tests ordered was not available in the NAMCS, we analyzed responses (yes/no) to the following question: "Was blood for the following laboratory tests drawn on the day of the visit or during the 12 months prior to the visit?" For the HbA1c test, adherence to guidelines was achieved if the response was "Yes." Admittedly, this definition was more lenient than the ADA guidelines, and may overestimate adherence. For the lipid profile test, which usually consists of tests for total cholesterol, high-density lipoprotein, low-density lipoprotein, and triglycerides, adherence to guidelines was achieved only if all four tests were conducted.

Continuity of care was measured by responses (yes/no) to the following question: "Are you the patient's primary care physician?" Patients were considered to have continuity of care if they were seen by their PCP. We also used the responses to distinguish patients who were seen by their PCPs from others who were seen by non-PCPs. We measured patients' age (as a continuous variable), sex, and race/ethnicity. The NAMCS coded a maximum of three provider's diagnoses in sequence according to the International Classification of Diseases, Ninth Revision, Clinical Modification, which we mapped to the Charlson comorbidity index [37] to calculate a comorbidity count for each patient.

#### Statistical analysis

We used the demographic and comorbidity profile of the patients to characterize primary care visits of patients with diabetes in 2013. The percentage of patients who received a HbA<sub>1c</sub> test or a lipid profile test in the preceding 12 months was calculated as an indicator of the degree to which physicians adhered to the guidelines. Pearson's chi-square tests were conducted for both HbA<sub>1c</sub> and lipid profile tests to examine whether PCPs more closely adhered to the guidelines than non-PCPs. The impact of continuity of care on achieving adherence for each test was estimated in logistic regression models with adjustment for patients' age, sex, race/ethnicity, and comorbidity count, and whether they were seen by their PCP. Visit weight and sampling strata variables were included in all the analyses to account for the 2013 NAMCS's design and sampling schemes and also to generate nationally representative estimates. All statistical analyses were conducted in SAS 9.4 (SAS Institute, Cary, NC, United States).

#### Results

In 2013, 17.0% of the primary care visits involved patients with diabetes, representing a total of 57 million visits (Table 1). Most patients in these visits were male, white, and non-Hispanic, with a mean age of 64 years and with few comorbidities except for diabetes. Patients were seen by PCPs in eight out of ten visits. Less than 60.0% of the patients with diabetes received recommended monitoring for diabetes (58.0% for HbA<sub>1c</sub> and 57.0% for lipid profile) during the preceding 12 months. PCPs ordered recommended tests more often than non-PCPs: 59.0% vs. 51.0%, respectively, for the HbA<sub>1c</sub> test ( $\chi^2 = 4.75$ , P = 0.03) and 59.0% vs. 44.0%, respectively, for the lipid profile test ( $\chi^2 = 12.69$ , P = 0.0004); see Table 2.

After we had accounted for patient characteristics, continuity of care was associated with improved guideline adherence (Table 3). Continuity of care was associated with increased odds of physicians adhering to guidelines for the HbA<sub>1c</sub> test (odds ratio 1.36, 95.0% confidence interval 0.98–1.88, P=0.06), but not significantly, and for the lipid profile test (odds ratio 1.76, 95.0% confidence interval 1.27–2.42, P=0.0006). Hispanic and female patients had significantly lower odds of receiving the HbA<sub>1c</sub> test and the lipid profile test, respectively. More comorbidities were associated with higher odds of receiving the HbA<sub>1c</sub> test but not the lipid profile test.



Weighted (%)

17

58

57

\_

47

53

63

15

18

4

85

Table 1. Characteristics of primary care visits of patien	ts with diabetes in 2013 in the Nation	al Ambulatory Medical Care
	Unweighted visit count	Weighted visit count
Full sample	2379	57,874,330
Adherence to ADA guidelines		
HbA <sub>1c</sub>	1363	32,218,643
Lipid profile	1262	31,227,035
Patient characteristics		
Age (mean)	64	-
Sex		
Male	1123	27,298,559
Female	1256	30,576,771
Race/ethnicity		
White	1732	36,610,996
Black	282	8,684,665
Hispanic	249	10,147,979
Other	116	2,430,691
Comorbidity (mean, Charlson comorbidity score)	0.7	-
Continuity of care	2009	49,164,753

Table 1. Characteristics Survey

Table 2. Weighted difference in ordering diabetes monitoring tests between primary care providers (PCPs) and non-PCPs

Test	PCPs	Non-PCPs	Difference	X <sup>2</sup>
HbA <sub>1c</sub>	59.0%	51.0%	8.0%	4.75 (P=0.03)
Lipid profile	59.0%	44.0%	15.0%	2.69 ( <i>P</i> =0.0004)

HbA<sub>1c</sub>, Hemoglobin A<sub>1c</sub>.

## Discussion

Diabetes is one of the five most common chronic diseases in the world and has substantial implications for the health and economic well-being of individuals, families, and nations [38]. Adherence to clinical guidelines by PCPs is key to effectively managing and preventing complications for those with diabetes, and is potentially more so for early detection and prevention of the condition for those at risk [39, 40].

Despite the widespread dissemination of clinical guidelines to both health care providers and patients [41], we found that PCPs' adherence to the ADA guidelines for monitoring diabetes was less than optimal in a nationally representative sample of primary care visits of patients with diabetes. While the proportion of patients receiving recommended tests in 2013 was much higher than in 2010 [21], there were still 40.0% of visits where patients with diabetes had not received any monitoring test in the preceding 12 months. This reveals a substantial gap between what was done in primary care settings and what is recommended for optimal diabetes management. It is worth noting that the estimates of physicians' adherence to diabetes guidelines derived from the NAMCS are lower than those from HEDIS measures [42], probably due to substantial differences in the sampling frame. For example, the HEDIS measure of HbA1c testing was calculated with all eligible patient encounters over the calendar year, while the NAMCS measure was based on PCPs' self-report of ambulatory visits in a random week with a retrospective timeframe of the 12 months before the visit. Without regularly monitoring glycemic and cholesterol levels, PCPs are unable to intensify treatment to prevent complications. Adhering to the guidelines for monitoring diabetes enables PCPs to make more timely decisions that are both clinically meaningful and cost-effective.

The current findings support the positive effect of continuity of care on diabetes quality of care [43]. Compared with prior studies that found no association between provider continuity

	ADA recommended test for monitoring diab								
		HbA <sub>1c</sub>			Lipid profile				
	OR	P value	95% CI	OR	P value	95% CI			
Continuity of care	1.36	0.06	0.98-1.88	1.76	0.0006	1.27-2.42			
Age	1.00	0.85	0.99-1.01	1.00	0.35	1.00-1.01			
Sex									
Male	-		-	-		-			
Female	0.78	0.04	0.62-1.00	0.73	0.008	0.58-0.92			
Race/ethnicity									
White	-		-	-		-			
Black	0.78	0.16	0.55-1.10	0.86	0.39	0.60-1.22			
Hispanic	0.69	0.04	0.48-0.99	0.77	0.17	0.53-1.11			
Other	1.94	0.03	1.05-3.59	1.20	0.51	0.69-2.09			
Comorbidity	1.36	0.004	1.11-1.67	0.91	0.38	0.74-1.12			

Table 3. Association of continuity of care and adherence to guidelines for monitoring diabetes.

ADA, American Diabetes Association; CI, confidence interval; HbA<sub>1c</sub>, hemoglobin A<sub>1c</sub>; OR, odds ratio.

and diabetes care, the continuity measure derived from the NAMCS was provider specific and based on provider self-identification as opposed to indexes that calculate a continuity score accounting for all providers of the patient or measures drawn from patient self-report [30, 31]. While answers to all of the identified challenges to adherence and implementation of guidelines for diabetes care are beyond the scope of this study, we found that continuity of care, a core tenet of primary care, increased the odds of adhering to guidelines, after patient characteristics had been controlled for [44-47]. These results reinforce the importance of the patient-provider relationship in management of a chronic disease. Last but not least, efforts to improve guideline adherence may yield larger benefits for Hispanic patients, who of all the race/ethnicity groups were the least likely to have been given a HbA<sub>1c</sub> test in the preceding 12 months, and among whom the prevalence of diabetes is still rising [1].

### Limitations

This study has several limitations. First, we were unable to measure the number of  $HbA_{1c}$  tests that were given in the preceding 12 months. Our definition of adherence for the  $HbA_{1c}$  test was more lenient than the ADA guidelines, which essentially lowered the threshold of being adherent. As a result, the 40.0% nonadherence rate might be an underestimate.

Second, also because of the unavailability of test counts, we were unable to determine whether overordering was an issue in nearly 60.0% of the visits where patients did receive HbA<sub>1c</sub> tests [48]. Third, responses in the NAMCS about whether tests were ordered for the patient in the preceding 12 months may be subject to recall bias. Last, our assessment of continuity of care was confined to interpersonal continuity. Future studies are encouraged to examine the influence of informational or administrative continuity on guideline adherence [49].

## Conclusions

Forty percent of primary care visits of patients with diabetes were missed opportunities where PCPs were not providing guideline-concordant monitoring tests, creating a substantial gap in the quality of care for diabetes in US primary care settings. Improving guideline-concordant disease monitoring in diabetes in primary care was highlighted by the World Health Organization to reduce the impact of diabetes worldwide [50]. Our findings suggest that enhancing patientprovider continuity of care may be effective in promoting guideline adherence.

## **Conflict of interest**

The authors have no conflict of interest.



# Funding

The American Board of Family Medicine Foundation supported Dr. Mainous.

# References

- Menke A, Casagrande S, Geiss L, Cowie CC. Prevalence of and trends in diabetes among adults in the United States, 1988–2012. J Am Med Assoc 2015;314(10):1021–9.
- Geiss LS, Wang J, Cheng YJ, Thompson TJ, Barker L, Li Y, et al. Prevalence and incidence trends for diagnosed diabetes among adults aged 20 to 79 years, United States, 1980–2012. J Am Med Assoc 2014;312(12):1218–26.
- Cowie CC, Rust KF, Ford ES, Eberhardt MS, Byrd-Holt DD, Li C, et al. Full accounting of diabetes and pre-diabetes in the U.S. population in 1988–1994 and 2005–2006. Diabetes Care 2009;32(2):287–94.
- Selvin E, Parrinello CM, Sacks DB, Coresh J. Trends in prevalence and control of diabetes in the United States, 1988–1994 and 1999–2010. Ann Intern Med 2014;160(8):517–26.
- Centers for Disease Control and Prevention. Diabetes data and trends: distribution of first-listed diagnoses among hospital discharges with diabetes as any listed diagnosis, adults age 18 years and older, United States, 2009. Atlanta: US Department of Health and Human Services; 2011.
- UK Prospective Diabetes Study Group. Intensive blood-glucose control with sulphonylureas or insulin compared with conventional treatment and risk of complications in patients with type 2 diabetes (UKPDS 33). UK Prospective Diabetes Study (UKPDS) Group. Lancet 1998;352(9131):837–53.
- UK Prospective Diabetes Study Group. Tight blood pressure control and risk of macrovascular and microvascular complications in type 2 diabetes: UKPDS 38. UK Prospective Diabetes Study Group. Br Med J 1998;317(7160):703–13.
- National Committee for Quality Assurance. HEDIS & performance measurement. 2016 [accessed 2017 May 1]. Available from: www.ncqa.org/hedis-quality-measurement.
- American Diabetes Association. Standards of medical care in diabetes 2013. Diabetes Care 2013;36(Suppl 1):S11–66.
- Tan E, Yang W, Pang B, Dai M, Loh FE, Hogan P. Geographic variation in antidiabetic agent adherence and glycemic control among patients with type 2 diabetes. J Manag Care Spec Pharm 2015;21(12):1195–202.
- Cheung BM, Ong KL, Cherny SS, Sham PC, Tso AW, Lam KS. Diabetes prevalence and therapeutic target achievement in the United States, 1999 to 2006. Am J Med 2009;122(5):443–53.

- Grant RW, Buse JB, Meigs JB. Quality of diabetes care in U.S. academic medical centers: low rates of medical regimen change. Diabetes Care 2005;28(2):337–442.
- Beaton SJ, Nag SS, Gunter MJ, Gleeson JM, Sajjan SS, Alexander CM. Adequacy of glycemic, lipid, and blood pressure management for patients with diabetes in a managed care setting. Diabetes Care 2004;27(3):694–8.
- Shojania KG, Ranji SR, McDonald KM, Grimshaw JM, Sundaram V, Rushakoff RJ, et al. Effects of quality improvement strategies for type 2 diabetes on glycemic control: a meta-regression analysis. J Am Med Assoc 2006;296(4):427–40.
- Dickerson LM, Ables AZ, Everett CJ, Mainous AG 3rd, McCutcheon AM, Bazaldua OV, et al. Measuring diabetes care in the National interdisciplinary primary care practicebased research network (NIPC-PBRN). Pharmacotherapy 2011;31(1):23–30.
- Morello CM, Zadvorny EB, Cording MA, Suemoto RT, Skog J, Harari A. Development and clinical outcomes of pharmacist-managed diabetes care clinics. Am J Health Syst Pharm 2006;63(14):1325–31.
- Irons BK, Seifert CF, Horton NA. Quality of care of a pharmacist-managed diabetes service compared to usual care in an indigent clinic. Diabetes Technol Ther 2008;10(3):220–6.
- Peterson LE, Blackburn B, Phillips RL, Puffer JC. Improving quality of care for diabetes through a maintenance of certification activity: family physicians' use of the chronic care model. J Contin Educ Health Prof 2014;34(1):47–55.
- Wisdom K, Fryzek JP, Havstad SL, Anderson RM, Dreiling MC, Tilley BC. Comparison of laboratory test frequency and test results between African-Americans and Caucasians with diabetes: opportunity for improvement. Findings from a large urban health maintenance organization. Diabetes Care 1997;20(6):971–7.
- Zoorob RJ, Mainous AG 3rd. Practice patterns of rural family physicians based on the American Diabetes Association standards of care. J Community Health 1996;21(3):175–82.
- Agency for Healthcare Research and Quality. The state of health care quality report: diabetes. Rockville: Agency for Healthcare Research and Quality; 2013.
- 22. Oude Wesselink SF, Lingsma HF, Robben PB, Mackenbach JP. Guideline adherence and health outcomes in diabetes mellitus type 2 patients: a cross-sectional study. BMC Health Serv Res 2015;15:22.
- Sidorenkov G, Haaijer-Ruskamp FM, de Zeeuw D, Denig P. A longitudinal study examining adherence to guidelines in diabetes care according to different definitions of adequacy and timeliness. PLoS One 2011;6(9):e24278.

- Bovier PA, Sebo P, Abetel G, George F, Stalder H. Adherence to recommended standards of diabetes care by Swiss primary care physicians. Swiss Med Wkly 2007;137(11–12):173–81.
- American Diabetes Association. Economic costs of diabetes in the U.S. in 2012. Diabetes Care 2013;36(4):1033–46.
- Stellefson M, Dipnarine K, Stopka C. The chronic care model and diabetes management in US primary care settings: a systematic review. Prev Chronic Dis 2013;10:E26.
- Goyder EC, McNally PG, Drucquer M, Spiers N, Botha JL. Shifting of care for diabetes from secondary to primary care, 1990–5: review of general practices. Br Med J 1998;316(7143):1505–6.
- Griffin S. Diabetes care in general practice: meta-analysis of randomised control trials. Br Med J 1998;317(7155):390–6.
- Koopman RJ, Mainous AG III, Baker R, Gill JM, Gilbert GE. Continuity of care and recognition of diabetes, hypertension, and hypercholesterolemia. Arch Intern Med 2003;163(11):1357–61.
- Mainous AG III, Koopman RJ, Gill JM, Baker R, Pearson WS. Relationship between continuity of care and diabetes control: evidence from the Third National Health and Nutrition Examination Survey. Am J Public Health 2004;94(1):66–70.
- Gill JM, Mainous AG III, Diamond JJ, Lenhard MJ. Impact of provider continuity on quality of care for persons with diabetes mellitus. Ann Fam Med 2003;1(3):162–70.
- National Center for Health Statistics. National Center for Health Statistics ambulatory health care data. 2012.
- National Center for Health Statistics. NAMCS 2012 micro-data file documentation. 2012.
- American Diabetes Association. Standards of medical care in diabetes 2012. Diabetes Care 2012;35(Suppl 1):S11–63.
- American Diabetes Association. Standards of medical care in diabetes 2011. Diabetes Care 2011;34(Suppl 1):S11–61.
- American Diabetes Association. Standards of medical care in diabetes 2010. Diabetes Care 2010;33(Suppl 1):S11–61.
- Charlson ME, Pompei P, Ales KL, MacKenzie CR. A new method of classifying prognostic comorbidity in longitudinal studies: development and validation. J Chronic Dis 1987;40(5):373–3.
- World Health Organization. Non-communicable diseases. 2017 [accessed 2016 May 1]. Available from: www.who.int/mediacentre/factsheets/fs355/en/.

- Mainous AG 3rd, Tanner RJ, Scuderi CB, Porter M, Carek PJ. Prediabetes screening and treatment in diabetes prevention: the impact of physician attitudes. J Am Board Fam Med 2016;29(6):663–71.
- Mainous AG 3rd, Tanner RJ, Baker R. Prediabetes diagnosis and treatment in primary care. J Am Board Fam Med 2016;29(2):283–5.
- Woolf SH, Grol R, Hutchinson A, Eccles M, Grimshaw J. Clinical guidelines: potential benefits, limitations, and harms of clinical guidelines. Br Med J 1999;318(7182):527–30.
- 42. National Committee for Quality Assurance. Comprehensive diabetes care. 2016 [accessed 2017 Sep 1]. Available from: www.ncqa.org/report-cards/health-plans/state-of-health-carequality/2016-table-of-contents/diabetes-care.
- 43. Francis MD, Julian KA, Wininger DA, Drake S, Bollman K, Nabors C, et al. Continuity clinic model and diabetic outcomes in internal medicine residencies: findings of the Educational Innovations Project Ambulatory Collaborative. J Grad Med Educ 2016;8(1):27–32.
- 44. Kenny SJ, Smith PJ, Goldschmid MG, Newman JM, Herman WH. Survey of physician practice behaviors related to diabetes mellitus in the U.S. physician adherence to consensus recommendations. Diabetes Care 1993;16(11):1507–10.
- Helseth LD, Susman JL, Crabtree BF, O'Connor PJ. Primary care physicians' perceptions of diabetes management. A balancing act. J Fam Pract 1999;48(1):37–42.
- Golin CE, DiMatteo MR, Gelberg L. The role of patient participation in the doctor visit. Implications for adherence to diabetes care. Diabetes Care 1996;19(10):1153–64.
- Anderson RM, Donnelly MB, Davis WK. Controversial beliefs about diabetes and its care. Diabetes Care 1992;15(7): 859–63.
- Pivovarov R, Albers DJ, Hripcsak G, Sepulveda JL, Elhadad N. Temporal trends of hemoglobin A1c testing. J Am Med Inform Assoc 2014;21(6):1038–44.
- Haggerty JL, Reid RJ, Freeman GK, Starfield BH, Adair CE, McKendry R. Continuity of care: a multidisciplinary review. Br Med J 2003;327(7425):1219–21.
- World Health Organization. Global report on diabetes. Geneva: World Health Organization; 2016.